Author’s response to reviews

Title: Are the stages of change relevant for the development and implementation of a web-based tailored alcohol intervention? A cross-sectional study

Authors:

Daniela N Schulz (DN.Schulz@maastrichtuniversity.nl)
Stef PJ Kremers (S.Kremers@maastrichtuniversity.nl)
Hein de Vries (Hein.deVries@maastrichtuniversity.nl)

Version: 2 Date: 21 November 2011

Author’s response to reviews:

Dear editors,

Please find enclosed our revised manuscript titled: Are the stages of change relevant for the development and implementation of a web-based tailored alcohol intervention? A cross-sectional study.

In the revised version, we included some information about the approval we received by the Medical Ethics Committee of Maastricht University and the University Hospital Maastricht (NL27235.068.09 / MEC 09-3-016) (page 6).

Yours sincerely,

Daniela Schulz (corresponding author)
Stef Kremers
Hein de Vries

CAPHRI/ Maastricht University
Faculty of Health, Medicine and Life Sciences
Department of Health Promotion
P.O. Box 616
6200 MD Maastricht
The Netherlands
T + 31 43 388 2832
DN.Schulz@maastrichtuniversity.nl